Author's response to reviews

Title: Tuberculosis screening and follow-up of asylum seekers in Norway. A cohort study.

Authors:

Ingunn IH Harstad (ingunn.harstad@ntnu.no)
Einar EH Heldal (einar.heldal@c2i.net)
Sigurd SLS Steinshamn (sigurd.steinshamn@ntnu.no)
Helge HG Garåsen (helge.garasen@ntnu.no)
Geir GWJ Jacobsen (geir.jacobsen@ntnu.no)

Version: 7 Date: 11 May 2009

Author's response to reviews:

Dear Madam/Sir

Thank you for accepting our manuscript in principle.

We hope we have done the revisions required under “major revisions” to your satisfaction. Under “minor revisions” we have not done any figure cropping and hope you will do them before publication. The other minor revisions have been done.

We have done some minor changes of words and spelling and have added one sentence under “data handling and analysis”.

If any demands for revisions have been misunderstood by us, we are happy to receive the manuscript again for further changes.